Author’s response to reviews

Title: In Vitro comparison of cyclic fatigue resistance of TruNatomy in single and double curvature canals compared with different nickel-titanium rotary instruments

Authors:

Amr Elnaghy (aelnaghy@mans.edu.eg)
Shaymaa Elsaka (shaymanaghy@mans.edu.eg)
Ayman Mandorah (amandorah@tudent.edu.sa)

Version: 2 Date: 26 Jan 2020

Author’s response to reviews:

Date: 26th January 2020
To: Dr McMillan, “BMC Oral Health”
From: "Amr Elnaghy", aelnaghy@mans.edu.eg
Subject: Submission OHEA-D-19-00580R1
Ref.: Ms. No. OHEA-D-19-00580R1
Title: In Vitro comparison of cyclic fatigue resistance of TruNatomy in single and double curvature canals compared with different nickel-titanium rotary instruments

Dear Dr. McMillan,

Many thanks for your message and the constructive comments to our paper OHEA-D-19-00580R1. We have read through the comments and our response to each of them is as stated below. The manuscript has been revised and uploaded as required.

Yours sincerely,

Amr Elnaghy

Comments:

Comment 1: Please upload your multi-panel figures as one figure. I.e. each separate figure (figure 1, figure 2, etc) should be one file, not multiple files.
Response: The separate figure (3) had been uploaded in one figure.

Comment 2: If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided before the Declarations.
Response: The abbreviations had been defined and provided before Declaration.

Comment 3: At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All
relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.
Response: A clean version of the manuscript had been uploaded.

Finally, many thanks to referees’ comments, efforts, and valuable suggestions regarding our paper. Also, many thanks to the Editor and all the Editorial Boards for their efforts.